Dear Editor,

Thank you very much for your professional advice and comments, which makes our manuscript more perfect and beautiful. After revising the manuscript according to the suggestions of the reviewers, we carried out the polishing of the manuscript, so the language of the manuscript has been adjusted and improved. Next, we will revert to the reviewer's suggestions one by one.

(1) Suggestions 2-7, 9, 11, 12, which belong to the syntax and details of the manuscript, have been revised and refined by us. See the manuscript in yellow for details.

(2) Suggestion 8, after discussion, we consider the latter part of the sentence to be repetitive, so it has been deleted.

(3) Suggestion 10, in the first two sentences of the Third paragraph, discusses the mild side effects of CCK-A agonists on the gastrointestinal tract. However, the remaining sentences state that CCK-A receptor antagonists have beneficial effects on the digestive system. Thus, the above studies suggest that research on the effectiveness and safety of targeting CCK-A receptors in the treatment of organic digestive diseases is still in the preliminary stage, and more in-depth exploration is required to provide a scientific basis for the prevention and treatment of these diseases and biliary dyskinesia in particular. See the manuscript in red for details.

Thanking you very much for considering our manuscript for potential publication. We are looking forward to hearing from you soon.

Signature ,